ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Filippini, Massimo; Srinivasan, Suchita

Working Paper Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam

Economics Working Paper Series, No. 20/338

Provided in Cooperation with: ETH Zurich, CER-ETH – Center of Economic Research

Suggested Citation: Filippini, Massimo; Srinivasan, Suchita (2020) : Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam, Economics Working Paper Series, No. 20/338, ETH Zurich, CER-ETH - Center of Economic Research, Zurich, https://doi.org/10.2020/othz.b.000424511

https://doi.org/10.3929/ethz-b-000424511

This Version is available at: https://hdl.handle.net/10419/222593

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

EHzürich

Research Collection



Working Paper

Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam

Author(s): Srinivasan, Suchita; Filippini, Massimo

Publication Date: 2020-07

Permanent Link: https://doi.org/10.3929/ethz-b-000424511 →

Rights / License: In Copyright - Non-Commercial Use Permitted →

This page was generated automatically upon download from the <u>ETH Zurich Research Collection</u>. For more information please consult the <u>Terms of use</u>.



CER-ETH – Center of Economic Research at ETH Zurich

Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam

M. Filippini, S. Srinivasan

Working Paper 20/338 July 2020

Economics Working Paper Series



Eidgenössische Technische Hochschule Zürich Swiss Federal Institute of Technology Zurich

Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam

Massimo Filippini^{1,2} and Suchita Srinivasan^{*1}

¹Center of Economic Research (CER-ETH), ETH Zürich, Switzerland ²Università della Svizzera italiana, Switzerland

Last revision: June 19, 2020

Abstract

Voluntary approaches to environmental policy can contribute to stemming environmental degradation in developing countries with weak institutions. We evaluate the role of a behavioral anomaly, limited attention paid by owners or managers, in explaining the voluntary adoption of environmental certification by small and medium enterprises (SMEs) in the food and beverage industry in Vietnam. We find that firms where owners or managers were inattentive were 30 percentage points less likely to receive environmental certification. Moreover, this effect is larger for firms that were previously inspected for technical violations, and that exported or bribed, and it is weaker for household enterprises.

JEL Classification: D22, D83; D91; O13; Q56; Q59

Keywords: Voluntary environmental standards; Limited attention; Small and medium enterprises; Food and beverage industry; Vietnam

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

^{*}Center of Economic Research (CER-ETH), ETH Zürich, Zürichbergstrasse 18, 8092 Zürich, Switzerland. Phone: +41 44 632 65 34, Fax: +41 44 632 10 50. <suchitas@ethz.ch>.

1 Introduction

Environmental degradation and climate change have significant negative consequences, not only on environmental quality and ecosystems, but also on food security, human health as well on livelihoods, both at the global as well as local levels (IPCC, 2014). Given the scale of the problem, especially in developing countries that are least equipped to deal with its consequences (World Bank, 2010), it is imperative to identify possible policy measures that may be relevant in ensuring environmental protection in these contexts, and to understand how to enhance their effectiveness.

Poor environmental quality is considered to be a hallmark of many areas in developing countries, especially as the process of development and urbanization gets underway. Pollution due to industrial activities, exploitation of natural resources, biodiversity destruction, deforestation, as well as the unsustainable development of agriculture and aquaculture systems are some of the typical problems encountered in these settings, often exacerbated by rapid population growth. The industrial sector is a significant contributor to environmental degradation in developing countries; for instance, it is one of the leading causes of water, solid waste and air pollution in Vietnam (International Trade Administration, 2019). Thus, it is critical to ensure that firms in developing countries find means to reduce their negative environmental impact, and thereby advance the cause of sustainable development.

The set of available policy instruments that may be utilized to mitigate the environmental impact of firms can be categorized as those primarily employed by the state, and others which require initiative to be taken by the firms themselves. The state must often intervene in the design and implementation of environmental policy instruments; these can broadly be classified as 'market-based instruments' that encourage changes in behavior through market signals (such as pollution charges, tradable permits, or subsidies) and 'non-market-based instruments' (including command-and-control regulations such as technology or performance standards, as well as information disclosure programs). Examples of environmental policy initiatives that can be taken by firms include corporate social responsibility (CSR) programs, as well as the adoption of voluntary certification (such as 'ecolabels') and environmental standards.

Voluntary certification and standards have been found to have the potential to achieve environmental targets in developing countries, at least under some conditions (Blackman 2010; Blackman and Guerrero 2012), in a context where traditional command-and-control policies have not always been very effective (Eskeland and Jimenez, 1992), often because of poor enforcement, lack of funding or trained personnel, lack of political will and a lack of awareness. Weak institutions, and the associated shortcomings in regulatory capacity and implementation, have meant that policy instruments such as voluntary standards and certifications may be an effective means to ensure 'self-regulation' by firms (Khanna and Liao (2014) provide a comprehensive overview of the literature on voluntary standards in developing countries).

The current literature on environmental voluntary standards for firms in developing countries has evaluated the motivation for firms to obtain these certifications, and found that firms often undertake these in response to pressure from regulators, consumers, investors, or environmental interest groups (Khanna and Liao, 2014). Examples of some firm-level characteristics that determine voluntary certification have been found to be foreign direct investment, firm size, whether the firm exports, and production of intermediate products (Tambunlertchai et al., 2013).

However, this literature also finds significant heterogeneity in adoption of these standards, and that not all firms that may benefit from their adoption actually do so (Khanna and Liao, 2014).

To the best of our knowledge, this literature has not considered owner (or manager)-specific behavioral traits that may determine voluntary certification outcomes. For instance, limited attention (or lack of knowledge) on the presence or availability of these standards could explain why some firms do not adopt them. Limited attention has been found to be an important determinant of firms not undertaking actions that may be in their best interest in developing countries (Beaman et al., 2014). The recent behavioral economics literature on limited attention suggests that firm owners may have trouble in attending to all aspects of their business, or have limited capacity to process all available information (Dellavigna 2009; Gabaix and Laibson 2006; Koszegi and Szeidl 2013; Hirshleifer and Teoh 2003). This may explain why in some contexts, firm owners may not adopt these standards, even if it may be beneficial to do so.

In this paper, our objective is to evaluate the role of limited attention in explaining the voluntary adoption of environmental certification by small and medium enterprises (SMEs) in the food and beverage industry in Vietnam. Vietnam has grown rapidly in recent decades, and small and medium enterprises have been the dynamic force behind this growth, comprising almost 98% of all enterprises, and employing more than 60% of the workforce (General Statistics Office, 2015). However, this industrialization has resulted in significant environmental deterioration: Vietnam ranks 132 out of 180 countries in the Environmental Protection Index (Wendling et al., 2018), and it ranks among the most affected countries in terms of air pollution (IQAir, 2019). The food and beverage industry is one of the most pollution-intensive industries in Vietnam, and it is significant for both its scale of production, and the pollution load.

At the same time, its growing middle class is also demanding products certified as being environmentally-friendly: according to a recent survey, Vietnamese consumers are among the most socially-conscious in Asia-Pacific. Up to 86% of consumers in Vietnam are willing to pay more for products and services from firms that are committed to having a positive social and environmental impact (Nielsen, 2015). Thus, in this context, not only may policy instruments such as voluntary environmental certificates contribute towards achieving environmental goals in Vietnam, but they are also likely to bring benefits to firms that obtain them (such as the potential for increased demand for their products).

In particular, in this study, we analyze the effect of limited attention on whether firms have applied for and received an environmental standards certificate, also known as the 'certification for environmental protection works'. We capture limited attention by whether owners or managers of the firm are unaware of (or lack knowledge on) the Law on Environmental Protection (LEP), which contains detailed and extensive information on how firms can obtain this certification, and is thus an important source of information on how firms can get certified. While obtaining this certification is mandatory for large firms belonging to specific sectors in Vietnam, in this paper, we focus on the voluntary adoption of these standards in a sample of small and medium enterprises. Firms can benefit from obtaining these certificates, by experiencing an increase in domestic demand for their products, as well as by increasing their likelihood of exporting their goods. Moreover, firms producing intermediate goods can use the standard to attract firms that require certification by their suppliers as potential customers. Of course, obtaining environmental certification is also likely to entail costs for these firms, which may be steeper for smaller businesses. However, it can be foreseen that at least for some firms, the benefits may outweigh the costs.

The focus of this paper is on a sample comprising predominantly small and medium enterprises (SMEs) in the food and beverage industry of Vietnam. We use unbalanced panel data from the UNU-WIDER Vietnam SME firm-level database (United Nations University UNU-WIDER, 2011) that collects information on about 2500 firms, mostly SMEs, from 2011-2015 biennially.

Our paper presents evidence spanning various estimation methodologies on the effect of limited attention on certification. We employ a probit estimator, a recursive bivariate estimator that is equipped to correct for possible endogeneity, and combine the recursive bivariate methodology with the Mundlak/Chamberlain adjustment (Mundlak 1978; Chamberlain 1982) to address possible unobserved heterogeneity. While the use of the recursive bivariate framework enables identification through assumptions on functional form, we also choose to employ an excluded instrument (the occupational skill level of the parents of the respondents) in our first-stage estimation.

Our contribution to the literature is that our study provides suggestive evidence on the role that behavioral anomalies (such as limited attention) can play in hindering voluntary certification by firms in a developing country. We use self-reported information on awareness about the LEP to establish that firms where owners or managers are more attentive to availability of instruments such as environmental certificates are more likely to make conscious efforts to treat environmental factors (and thus satisfy environmental standards).

Moreover, we also add to the literature on the determinants of limited attention on part of owners and managers, and identify both firm-level and respondent-level factors that may lead to individuals being more aware/knowledgeable of environmental laws. Of course, these factors are likely to be context-specific, however, they still provide interesting insights into how policy-makers can target information and education campaigns.

The policy implications of this study relate to the importance of informing and educating firm owners and managers about the availability of voluntary environmental instruments, as well as enabling them to understand laws better, given that awareness about the laws is likely to weaken at least some information constraints for small and medium enterprises. For instance, it can be expected that firms that are attentive to the laws are not only going to comply with mandatory environmental regulations, but as our results suggest, may also be more likely to adopt voluntary instruments. This is of pivotal interest in developing countries, where weak institutions and a lack of information may hamper the uptake of environmental policy initiatives by firms. Thus, bringing laws and regulations to salience for firms is a low-hanging fruit that policy-makers can effectively utilise to improve environmental outcomes.

The structure of the paper is as follows: Section 2 includes a background on the environmental law and voluntary standards in Vietnam, Section 3 provides a brief review of the literature, Section 4 provides details on the data and methodology used for the analysis, Section 5 includes the main results of the paper as well as additional results, while Section 6 concludes and includes policy implications.

2 Background

Vietnam is one of the most rapidly transforming countries not only in East Asia, but also in the world: following the phase of economic and political reforms called Doi Moi in 1986,

the economy liberalised significantly, and poverty rates declined sharply. For instance, from 2002 to 2018, poverty rates declined from over 70% to below 6% (at USD 3.2 per day at purchasing power parity), and gross domestic product per capita increased 2.5 times, to USD 2,500 in 2018 (World Bank, 2019). However, future expected economic growth, coupled with the associated industrialization and urbanization, imply that there may be significant pressures on environmental sustainability in the country.

According to estimates from the General Statistics Office (GSO) of Vietnam, almost 98% of all enterprises operating in Vietnam were SMEs (General Statistics Office, 2015). As of 2015, they accounted for about 64% of employment, and 45% of the country's GDP. Given their share in Vietnam's economy, and the contribution of the industrial sector to environmental degradation in Vietnam, it is reasonably important to ensure that owners and managers of SMEs are environmentally-conscious, and that they proactively address environmental issues.

The key regulatory instrument at the disposal of policy-makers in Vietnam is the LEP. The framing of environmental regulation in Vietnam began with the passing of the LEP in 1993, providing the framework for legal, institutional and administrative instruments for environmental protection in Vietnam. The law was substantially amended in 2005 (as well as in 2015) to increase the stringency of enforcement, as well as to increase the scope of regulation to tackle the environmental challenges in the country. This law serves as the basis for industrial pollution management policies in Vietnam, as well as for the framing and implementation of environmental standards.

The LEP requires that firms that can produce at a certain minimum scale in specific industries conduct an environmental impact assessment (EIA) before starting operations, with the right to start or continue their business only when their EIA reports have been reviewed, and approved by the State Management Agency for Environmental Protection (The Government of Vietnam, 2015; Vietnam Law and Legal Forum, 2008).¹ The EIA is expected to provide a detailed description of the project, as well as important information on its environmental impact on the site, and potential risks due to environmental incidents.

On being granted the approval for the EIA, they then must submit documents in order to get an inspection done for pollution control. Firms are granted an environmental standards certificate (ESC), or "certificate for completion of environmental protection works" (as it is now called after the amendment of the LEP in 2015) if they comply with the requirements set forth in the EIA, specifying the environmental factor that firm owners are aiming to treat, as well as successfully pass the inspection requirements (Vietnam Law and Legal Forum, 2008).

While obtaining the ESC is mandatory for firms having a minimum production capacity in some industries,² other firms can obtain the ESC on a voluntary basis.³ In this paper, we will focus on the voluntary adoption of the ESC by firms in the food and beverage industry. The relevant legal documents contain information on the requirements, not only for mandatory compliance by firms that are required to conduct the EIA and obtain the ESC, but also for firms that are

¹All other firms that are not required to submit an EIA (i.e. the smaller firms) are expected to submit an environmental protection plan, barring firms belonging to specific sectors or lines of business, such as retail establishments.

²Firms that do not conduct an EIA (if they are required to do so), and thus do not obtain the ESC, may face sanctions, varying in severity from monetary penalties, to a complete shutdown.

³The Government of Vietnam (2015) provide the list of sub-sectors within each industry that are required to obtain the ESC, aw well as the associated production capacities for each sub-sector.

looking to obtain voluntary certification. Thus, being knowledgeable or aware of the LEP is likely to be a pivotal channel of information on the steps to obtain voluntary certification.

The food and beverage industry comprised the largest share of all enterprises, with 4480, or 21.9% of 20444 manufacturing firms in the country in 2004 belonging to it (Dore et al., 2008). Not only did this sector represent 40.4% of the total value produced by locally owned state enterprises, it also represented the largest share of all industrial value produced by both the domestic private sector (29.8%), and the foreign-invested sector (18.7%) in Vietnam in 2004 (Dore et al., 2008). Thus, it is an economically significant sector in Vietnam.

The food and beverage industry is also one of the most pollution-intensive industries in Vietnam. Food processing (particularly, seafood processing)⁴ as a sub-industry has a very high overall environmental impact, and is significant for both its scale of production, and the pollution load. Dore et al. (2008) identify this sector, along with textiles, and chemical and chemical products, as the three biggest contributors to industrial pollution in Vietnam.⁵ Food processing sub-industries are also some of the biggest contributors to the release of toxic materials into the land in Vietnam (Dore et al., 2008).

3 Previous Literature

Our paper fits into two strands of the economic literature, one on the determinants of adoption of voluntary environmental standards and certificates by firms, and the other on the role of limited attention as a behavioral anomaly in determining economic decisions.

Voluntary approaches for environmental protection have been used in several settings as an environmental policy instrument. In general, they have gained traction in industrialized countries, because of the benefits they offer over command-and-control regulation (such as speedier implementation, or reducing the administrative burden on regulatory agencies). Common examples of voluntary approaches adopted include participation in public voluntary programs established by regulatory agencies, negotiated agreements between governments and firms, and initiatives adopted independently by firms, such as certification, eco-labels, and adoption of environmental management systems (such as the ISO-14001) (Khanna and Liao, 2014).

The literature on the adoption of voluntary environmental certification and standards in these countries has found that the effectiveness of this policy is, in many cases, mitigated due to "selection-type" problems, namely that only the cleanest firms and plants obtain these certificates (Vidovic and Khanna, 2007). For these firms, the marginal costs of certification are low, and the benefits are often of considerable magnitude.

Firms often adopt these standards under the influence of regulators, consumers, or environmental interest groups. They may do so to minimize current and future costs of compliance with

⁴Seafood processing includes cold-storage; freezing; drying; and smoking and canning of fish, shrimp, squid, shellfish, algae, and other marine products. It also includes the production of fish ol and sauces, seasoning products, and fish meal.

⁵While there is scant information on pollution abatement and control expenditures by firms in Vietnam, the Vietnam General Statistics Office (GSO) conducted a business survey in 2002 where this was asked to firms. While not all firms provided this information, among those that did, the food and beverage industry was the leading industrial sector for pollution abatement and control expenditures (in terms of amount spent) (Dore et al., 2008).

environmental regulations (as Decker (2003) shows, firms obtain permits for new facilities more quickly if they have engaged in voluntary abatement); they may do so to nudge regulators about their compliance (and thus, avoid stricter monitoring), or they may do so to influence the design and prevent the increase in stringency of future regulations (Segerson and Miceli, 1998). For instance, Videras and Alberini (2000), Innes and Sam (2008) and Vidovic and Khanna (2007) all find that firms were more likely to join the voluntary 33/50 Program of the United States Environmental Protection Agency (EPA) if they had more polluting sites, and released more toxic chemicals.

Of course, signalling about the environmentally-friendly nature of their production to consumers (Arora and Gangopadhyay, 1995), or attracting workers (who may be drawn to firms that project themselves to be environmentally conscious) are other important reasons for firms in industrialized countries to get certified on a voluntary basis (Khanna and Liao, 2014). Evidence from industrialized countries also suggests that firms with more educated managers or workers, or those where managers had a bent towards protecting the environment, were more likely to adopt voluntary certificates (Ervin et al. 2012; Khanna and Speir 2013).

This literature finds a role for mandatory environmental regulation in spurring firms to obtain voluntary certification: Arimura et al. (2008) use data on Japanese firms, and find that plants subject to environmental performance standards and input taxes were more likely to be certified with the ISO 14001, while Potoski and Prakash (2005) find that plants inspected more often were more likely to be certified with it. Other important determinants of certification for firms in industrialized countries include firm size (which is found to be positively associated with adoption) (Arimura et al. 2008; King et al. 2005; Potoski and Prakash 2005) and sales to foreign buyers (King et al., 2005).

Environmental regulation take on a different dimension in developing countries, where weak institutions are a deterrent towards enforcement of regulation. Several papers in development economics have shown that in these settings, policy failures are common, especially when weak incentives and corruption are rampant (Banerjee et al. 2008; Duflo et al. 2012; Banerjee et al. 2013). Eskeland and Jimenez (1992) provide an introduction to the problems of enforcing policy instruments such as fiscal incentives and performance standards in settings with weak institutional characteristics.

The literature on voluntary environmental certification as an environmental policy instrument in developing countries is relatively thin. Given weak institutions, limited capacity to regulate and costly monitoring, the case for voluntary environmental programs such as certification and standards in developing countries is strong. Command-and-control regulation has been known to face institutional and political challenges in low and income country-settings (Eskeland and Jimenez, 1992). However, Vincent (2010) provides a comprehensive summary of studies that have found that alternate pollution control measures (such as public disclosure and voluntary programs) may have limited success in developing countries in some contexts. In contrast to industrialized countries, voluntary programs in developing countries are generally used to facilitate compliance with mandatory regulation (Blackman and Guerrero, 2012).

Blackman et al. (2010) evaluated Mexico's Clean Industry Program, and found that plants that were fined for regulatory violations were more likely to participate in this voluntary program (that awarded firms recognition for submitting to an environmental audit), and that these firms were large and exported to foreign markets (and thus, were more likely to work to satisfy their

customer's requirements, rather than only meet domestic regulations). The finding regarding firm size determining participation in voluntary programs is confirmed by Christmann and Taylor (2001) for Chinese firms and Tambunlertchai et al. (2013) for Thai firms, and regarding firms linked to overseas markets being more likely to participate is also confirmed by Montiel and Husted (2009) for Mexican firms, and again by Tambunlertchai et al. (2013).

Blackman et al. (2010) suggests that there are disparities in the adoption of voluntary programs, both within and across developing countries. While several firm-specific characteristics have been used to explain their adoption, the role of owner-specific traits, such as behavioral anomalies or biases that they may have, has not been explored in the literature, to the best of our knowledge. One factor that may explain the under-adoption of such instruments by some firms is a lack of knowledge or awareness of the availability of such instruments, or limited attention paid to them by owners of the firms. The vast theoretical literature on limited attention (Dellavigna 2009; Gabaix and Laibson 2006, Koszegi and Szeidl 2013; Bordalo et al. 2013) pinpoints that owners and managers at firms may have limited ability to process all available information, and thus that they may not always take optimal business decisions.

There is an empirical literature from industrialized countries on the role of limited attention (also known as behavioral inattention) in undermining consumer decisions (Chetty et al. 2009; Hossain and Morgan 2006). There is also emerging evidence from developing countries: limited attention has been found to be an important determinant of sub-optimal decisions made by firms, as well as individuals. Beaman et al. (2014), for example, provide evidence from two randomized controlled trials with micro-enterprises in Kenya on the impact of bringing to salience the importance of keeping change, and found that firms were more attentive to keeping change to run their businesses in response to being informed about it. Hanna et al. (2014) apply the model of limited attention to explain the decision of seaweed farmers in Indonesia to ignore information on pod size, which is an important input in determining output. Karlan et al. (2016) provide evidence from Bolivia, Peru and the Philippines on the role of reminders in inducing saving behavior among clients of banks.

To our knowledge, our paper is one of the first economic studies to look at the effect of limited attention (or knowledge) on voluntary adoption of environmental certificates by firms in a developing country. We are aware of a couple of other studies that have looked at the role of policy or regulation awareness in environmentally-oriented settings: Nkonya et al. (2008) assessed the effect of awareness on compliance with by-laws related to community resource management in Uganda, and Cerruti et al. (2019) evaluated the impact of policy awareness on the decision of households to invest in energy-efficient cars in Switzerland. Both studies have found awareness of regulations/policies to have a significant impact on decisions.

While we confirm some of the previous results on the role of firm size and exporting behavior on the decision of firms to adopt environmental certificates voluntarily in developing countries, the novelty of our study is that we focus on the role of limited attention paid to the LEP, which is a source of information on these certificates, in determining this outcome. Limited attention in this setting may imply a lack of awareness on the existence of the law. Another aspect of limited attention may be a lack of understanding of what the law entails for the firms. While we are not able to disentangle the two effects in our data, we are able to capture the aggregate effect of a lack of knowledge, or awareness, on adoption of these certificates. We control for characteristics that may determine the mandatory adoption of these standards, as well as for inspections faced by firms (one of the means of enforcement), and thus we focus on the role that limited attention may play in their voluntary adoption.

4 Empirical Approach and Data

4.1 Model Specification and Empirical Approach

In order to estimate the effect of limited attention on the likelihood of obtaining the certificate for meeting environmental standards on a voluntary basis, we first estimate a simple discretechoice model. Given the binary nature of our dependent variable, from an econometric point of view, we choose to estimate a probit model. The model specification is as follows:

$$C_{i,t} = \alpha_0 + A_{i,t}\alpha_1 + X_{i,t}\alpha_2 + \eta_j + \gamma_t + \lambda_k + \mu_{i,t}$$

$$\tag{1}$$

We define the variables used in the model below. $C_{i,t}$, our dependent variable, is a binary variable capturing whether the firm 'i' had an ESC (or a certificate for satisfying environmental standards) in period 't'. The main independent variable $A_{i,t}$ is a binary variable, indicating whether the respondent (either the owner, or the manager of the firm) had paid 'limited attention' to the Law of Environmental Protection (LEP) (which provides detailed information on how firms can obtain the ESC) in period 't'.⁶ $X_{i,t}$ denote the set of controls, η_j denotes a set of province 'j'-specific dummy variables, γ_t denotes year fixed effects, and λ_k denotes sub-sector 'k'-specific dummies. The set of control variables in the estimation $X_{i,t}$ includes various firm-level characteristics (such as whether the firm is a household enterprise, whether the firm has internet access, age of the firm, whether the firm exports, etc.) and respondent-level characteristics (such as gender, age, level of education, etc.).

As mentioned in the previous section, the adoption of the ESC is mandatory for large firms (based on annual production capacity) within some sub-sectors, and it is voluntary for other firms within the food and beverage industry. In this study, we focus on the voluntary adoption of these standards, accounting for the firms that need to obtain the ESC on a compulsory basis by including a dummy variable that captures compliance. More details are provided in the Data sub-section.

In our dataset, the LEP awareness variable is categorical and with four levels for possible responses: "good", "average", "poor" and "no knowledge or not of my interest". There are two reasons for us choosing to convert this information into a binary variable. Firstly, the category "no knowledge or not of my interest" has the maximum responses, with 59% of respondents belonging to it (the shares are 2%, 13% and 26% for the remaining three categories, respectively). Secondly, the differences between the categories "good, "average" and "poor" are not necessarily clearly defined, and may introduce an element of subjectivity into our measure, whereas the fourth category is more unambiguous in definition, in our opinion. Thus, we categorize firms as having 'limited attention' of the LEP if the respondents stated that they had no knowledge of the law, or that it was not of their interest.

⁶While the 2005 Law on Environmental Protection would have been valid for the first two years of our study (2011 and 2013), the 2015 amendment would be relevant for the firms in the final year of our data (2015). However, both versions of the law are very similar in aspects related to obtaining the EIA approval and the ESC, and thus are treated as identical in our analysis.

While this estimation is a natural starting point, it is difficult to rule out the concern of endogeneity in the estimation of model 1. It is highly plausible, for example, that firms that have the certificate for meeting environmental standards are more likely to be aware of the LEP, which suggests possible reverse causality in the above estimation. Moreover, it is likely that both the likelihood of having received a certificate, as well as awareness of the LEP, are influenced by factors unobservable to an econometrician such as respondent-level characteristics (for instance, how law-abiding they are), or firm-level traits (such as whether the firm hired the services of a legal consultant), leading to the possibility of correlated unobservables. Lastly, measurement error in the variable capturing limited attention is also a possible threat to identification, since this information is self-reported.

In order to address these concerns, we estimate a bivariate probit model, adopting the recursive structure proposed by Maddala (1986). This is our preferred choice of estimation to mitigate endogeneity concerns, given that we use a binary version of the possibly endogenous variable in our model, following Greene (2008, 2018). In this context, we model the decision to get certified as a two-stage decision process. In the first stage we model whether the respondent (owner or manager) had limited knowledge (or paid limited attention to) the LEP, and then in the second stage, we model the choice of the firm to get the environmental certification. This two-stage process is modelled as shown below:

$$A_{i,t}^{*} = \beta_{0} + \beta_{2}Z_{i,t} + \beta_{3}X_{i,t} + \eta_{j} + \gamma_{t} + \lambda_{k} + \epsilon_{i,t}, A_{i,t} = 1 \text{ if } A_{i,t}^{*} > 0, A_{i,t} = 0 \text{ otherwise,}$$

$$C_{i,t}^{*} = \alpha_{0} + \alpha_{1}A_{i,t} + \alpha_{2}X_{i,t} + \eta_{j} + \gamma_{t} + \lambda_{k} + \mu_{i,t}, C_{i,t} = 1 \text{ if } C_{i,t}^{*} > 0, C_{i,t} = 0 \text{ otherwise.}$$
(2)

where the assumption is that

$$[\epsilon_{i,t}, \mu_{i,t}] \sim \phi_2[(0,0), (1,1), \rho], \rho \in [-1,1]$$
(3)

and where $A_{i,t}^*$ and $C_{i,t}^*$ denote vectors of the latent continuous variables (instead of which the binary variables $A_{i,t}$ and $C_{i,t}$ are observed to the econometrician), $Z_{i,t}$ denotes the vector of our instrumental variable, $X_{i,t}$ is the vector of control variables, and $(\epsilon_{i,t},\mu_{i,t})'$ is a vector of error terms described by ϕ_2 , the bivariate standard normal distribution, with correlation given by ρ . The remaining variables are defined as they were earlier.

Model 2 is generally identified, even if $X_{1i} = X_{2i} = X_i$, i.e. the same set of control variables are used in both stages, granted enough variation is provided by the exogenous covariates in the model. In the narrow case in which $X_{1i} = X_{2i} = a$ set of dummy variables, the absence of an exclusion restriction may result in a failure of identification. A detailed discussion of this case, and on identification in the recursive bivariate model more generally, may be found in Wilde (2000), Mourifié and Méango (2014) and Han and Vytlacil (2017). Further discussion on identification is beyond the scope of this paper.

However, we choose to incorporate an excluded instrument in our model $(Z_{i,t})$, which is a constructed measure of the "skill level" of the occupations pursued by the parents of the respondent. It is an index, generated as the sum of the levels of skills of each parent, and measured using a categorical variable. We use this as an instrumental variable under the

premise that respondents (owners or managers) who are children of individuals who worked in high-skilled jobs (who are thus also more likely to have a higher level of education) are more likely to be aware of laws in general. Moreover, it is unlikely that the skills of the parents of the respondent will directly influence the certification received by the firm, once we control for respondent-level characteristics such as age, education, etc.⁷

In order to control for unobserved heterogeneity across firms in our sample, we could use a fixed effects version of the bivariate probit model. However, as discussed in Greene and Zhang (2019), the use of a fixed effects model within the framework of a nonlinear panel data model gives rise to the 'incidental parameter problem' and thus the estimates would be biased. Therefore, in order to deal with time-invariant unobserved heterogeneity, we choose to adopt the correlated random effects approach suggested by Mundlak (1978), Chamberlain (1982), and Wooldridge (2005, 2010). This approach is based on including time-averages of all variables having positive within variation in both first and second stages of the model (2) as controls. The model that we estimate is then transformed as follows:

$$a_{i,t}^{*} = \beta_{0} + \beta_{2} z_{i,t} + \beta_{3} x_{i,t} + u_{i} + \eta_{j} + \gamma_{t} + \lambda_{k} + \epsilon_{i,t}, a_{i,t} = 1 \text{ if } a_{i,t}^{*} > 0, a_{i,t} = 0 \text{ otherwise,}$$

$$c_{i,t}^{*} = \alpha_{0} + \alpha_{1} a_{i,t} + \alpha_{2} x_{i,t} + u_{i} + \eta_{j} + \gamma_{t} + \lambda_{k} + \mu_{i,t}, c_{i,t} = 1 \text{ if } c_{i,t}^{*} > 0, c_{i,t} = 0 \text{ otherwise.}$$
(4)

where we now assume that u_i can be substituted as

$$u_i = \psi_1 + \bar{x}_i \psi_2 + \delta_i, \delta_i | x_i \sim Normal(0, \sigma_{\delta_i}^2)$$
(5)

The Mundlak-Chamberlain adjustment models the firm-level heterogeneity term (u_i) to be a linear function of all exogenous variables x_i , including those that are omitted from model (2) above. In equation 5, the term δ_i is assumed to be independent of x_i . By substituting equation 5 in equation 4, $c_{i,t}$ is allowed to be correlated with unobserved heterogeneity, and the time-varying omitted factor (this approach has been extensively summarized in Greene (2018) and in Wooldridge (2010)).

Summarizing, our approach in this paper is to adopt three methodologies to evaluate the impact of limited attention on the likelihood of a firm having received the ESC: the probit model, the recursive bivariate probit model, as well as the recursive bivariate probit model with the Mundlak/Chamberlain adjustment. In the next section, we describe our data, and provide some important summary statistics on our modelling variables.

⁷In our sample, the average age of the owners/managers of the firms is 48 years. Within the household of the respondents, the share of adults over the age of 60 is only about 9.5%, with 71% of respondents stating that their household had no members over the age of 60. Thus, we are of the opinion that it is unlikely that parents of the respondents will directly influence certification decisions, and that the magnitude of this effect, if it exists, will be negligible.

4.2 Data

For this study, we use data from the UNU-WIDER Vietnam SME firm-level database (United Nations University UNU-WIDER, 2011). The database tracks a sample of 2500 predominantly small and medium-sized firms in nine provinces of Vietnam biennially over the period 2011-2015, creating an unbalanced panel. The data set collects information on the economic accounts, as well as data on various enterprise-level, as well as some employee-level characteristics. The enterprises surveyed are distributed over approximately 18 different industries, including the food and beverage. Firms are classified according to the current World Bank definition, with micro-enterprises having up to 10 employees, small-scale enterprises up to 50 employees, medium-sized enterprises up to 300 employees, and large enterprises having more than 300 employees.⁸ The database also includes variables related to firm performance, enterprise history, employment, business environment, and owner/ manager background characteristics.

The regression sample for our main results includes firm in the food and beverage industry with manufacturing as the main sector that do not change their industry over the duration of the sample. After accounting for missing values for some variables, this sample comprises 2216 observations. In order to focus on the voluntary adoption of the ESC in our sample, we first seek to identify those firms that are required to obtain the ESC on a mandatory basis according to the LEP, and account for this in our estimations by including a dummy variable that takes the value 1 for these firms that are required to comply, and thus obtain the ESC. The two main determinants for mandatory compliance by firms are the sub-sector that they operate in, and the production capacity of the firm. The appendix to The Government of Vietnam (2015) provides this list of sub-sectors, as well as the production capacity thresholds. Firms that belong to specific sub-sectors, and have a production capacity larger than this predefined threshold, are required to obtain the ESC.

From our data set, we are able to ascertain the sub-sector within which a firm operates (identified by the four-digit Vietnam Standard Industrial Classification (VSIC) code; the exhaustive list of the VSIC codes can be found in Prime Minister's Office (2018)), as well as the quantity produced by a firm in the last year (for its top three products). We use this information on the quantity (of its primary good) produced by the firm in the last year to identify whether a firm meets the threshold for production capacity that is relevant for a firm in that sub-sector to obtain the ESC. If a firm produces more than this threshold in even a single year (out of a maximum of three years), we code this dummy variable as taking the value 1.⁹

We are able to approximately match almost all the sub-sectors in our data set with those listed in The Government of Vietnam (2015), except one sub-sector, namely "manufacturing of other food products: n.e.c", which comprises about 17% of the observations in our regression sample. We were not able to find a corresponding match for this sub-sector, and thus unable to determine the threshold. To be conservative, we choose to code the dummy for compliance

⁸The maximum value of the total labour force among the 2216 observations in our main estimation sample is 280.

⁹While we have information on the quantities and units of production for most firms, we were not able to translate the units into English for some of the observations. In other cases, we had to make assumptions to convert the units mentioned in the database to those used to define thresholds in (The Government of Vietnam, 2015). For instance, some firms producing beverages in our data set reported production in terms of boxes of product, while the threshold is defined in terms of litres. In such cases, we made suitable assumptions to identify which firms may meet the thresholds.

as one for all large firms in this sub-sector, based on the size of their total labour force. Thus, we code this dummy as 1 if a firm belonging to this sub-sector employs more than 6 workers (which is the median number of workers in our data sample over all firms).¹⁰ The dummy variable takes the value of '1' for about 17% of the observations in our regression sample (for 379 out of 2216 observations). We are also able to confirm the main results of the paper (provided in columns (1) and (3) of Table 3) by dropping all observations for which this dummy takes the value 1 (results can be provided on request).¹¹

To capture any residual effects on adoption of the ESC due to mandatory compliance by firms, in all our estimations, we include dummy variables for each sub-sector within the food and beverage industry where firms are required to obtain the ESC. In order to control for firm size, we include the total labour force of the firm as a control variable in all regressions.

The geographical coverage of our study is nine provinces of Vietnam, from three different regions, the north (Hanoi, Phu Tho, and Hai Phong), south (Ho Chi Minh City, Long An, and Khanh Hoa), and central (Nghe An, Quang Nam, and Lam Dong), including some of the most important manufacturing centers of the country (such as Ha Noi, Hai Phong, Quang Nam, Ho Chi Minh City, and Long An). The survey is representative at the province level (Sharma and Tarp, 2018). In this paper, we focus our analyses on the food and beverage industry, which is one of the most important industries in Vietnam, and also the largest in terms of sample size in our data set (it comprises 34% of the total observations).

Table 1 below provides summary statistics on the main explanatory variables, as well as on the dependent variable, for the regression sample. While about 17% of the firm-year observations suggest having received the certification for meeting environmental standards, about 59% of respondents in our data sample said that they had no knowledge of the LEP, or didn't care about it, i.e. paid limited attention to it. About 77% of the respondents were owners, and 51% of them were male. The average age of the respondent is about 48 years in our sample. However, the share of respondents who were college-educated was fairly low, at about 19%. About 83% of respondents rely on the income or profits from the firm as their main source of household income.

Importantly, our data set also contains information on whether the firm was inspected for various reasons (technical reasons such as environmental compliance or for fire-safety, policy-related reasons such as labor code or tax law violations, or other reasons, such as after accidents). We control for enforcement, by using a lagged indicator of the total number of technical inspections. Specifically, our variable captures the median of the lag of number of technical inspections over all firms other than firm 'i' at the district level. Thus, we create a measure of the average number of lagged inspections at the district level for all firms excluding the firm in question, so as to avert endogeneity concerns, and ensure that we control for the monitoring efforts. We find that the mean of this variable is 0.13 for our regression sample.¹²

¹⁰We also used other values for the size of labour force to determine this cut-off, and were able to confirm the results of at least the probit estimation in column (1) of Table 3 for all other values. We also attempted dropping this sub-sector from our analysis, and in this case we were also able to confirm the probit results of column (1) of Table 3. While we also tried estimating the recursive bivariate models in columns (2) and (3) of Table 3 using varying cut-offs, as well as by dropping this sub-sector from our data set, we were unable to achieve convergence in these estimations.

 $^{^{11}}$ We were not able to confirm the results of column (2) of Table 3, due to a lack of convergence of the model.

¹²While taking the lag of the number of inspections would have implied a significant loss in observations,

Table 1: Summary statistics for the regression sample

Variable	Mean	Std. Dev.	Minimum	Maximum
Certification for meeting environmental standards	0.167	0.373	0	1
Paid limited attention to LEP	0.591	0.492	0	1
Whether respondent is owner	0.773	0.419	0	1
Whether respondent is male	0.509	0.500	0	1
Age of the respondent	47.666	10.666	17	89
Whether respondent is college educated	0.185	0.388	0	1
Whether profit/income from firm is the main source of household income	0.829	0.376	0	1
Median number of inspections at the district level	0.132	0.402	0	2
Age of the firm	16.768	10.563	2	76
Whether firm is a household enterprise	0.825	0.380	0	1
Total labor force of the firm	8.843	20.520	1	280
Whether firm has certificate of land use right	0.799	0.401	0	1
Whether firm has multiple owners	0.072	0.259	0	1
Whether firm has internet access	0.188	0.391	0	1
Number of personal computers owned	0.755	3.053	0	98
Current (within district) market share	37.369	42.398	0	100
Whether firm has ever paid a bribe	0.305	0.461	0	1
Whether firm exports	0.035	0.184	0	1
Whether firm produces more than one good	0.069	0.254	0	1
Whether firm faces competition in its field of activity	0.839	0.368	0	1
Whether firm had labor disputes in the last two years	0.001	0.037	0	1

Notes: Summary statistics are calculated for the regression sample of columns (2) and (3) of Table 3, using 2216 observations. This regression sample includes firms in the food and beverage industry that did not change either location or industry of operation over the three years of the sample (2011, 2013 and 2015) and had manufacturing as the main production sector.

The average age of the firm in our sample is about 17 years, suggesting that these firms are not very young, on average. About 82% of these businesses are run as household firms, with an average total labor force size of about 9 workers per firm. 80% of firms have the certificate for land use right, namely the right to use the land (since private ownership of land is forbidden in Vietnam). About 93% of firms have a single owner, while about 19% of firms have access to the internet, with firms owning about one computer on average. The mean district-level market share of these firms is about 37%.

About 31% of the respondents admitted to having paid bribes for some reason related to the business. Corruption and bribery, especially in the form of "grease money" to fulfil basic tasks and services, is rampant in Vietnam (Bai et al., 2019). While it is unknown whether the firms in our sample paid bribes in order to get the certification, or to avoid getting it, it is likely to be a determinant of the certification process for firms in Vietnam. About 4% of firm-year observations consist of firms that export, while about 7% comprised firms that had diversified their product base (i.e. produced more than one good). A very large share of firms (about 84%) said that they faced competition in their field of activity. About 0.14% of respondents admitted to have had a labor dispute within the firm in the past two years.

Table 2 presents some additional statistics on levels of certification for environmental standards, as well as on the levels of awareness on the LEP for the respondents. The average certification

given that we would have no values for this variable for 2011, we were able to link our data set to a previous version of this survey which was conducted in 2009, and collected information on inspections. We were able to match most firms in our 2011 sample to the 2009 data set, and thus, we do not end up losing a significant number of observations (out of a total of 748 firms in the food and beverage industry in 2011, we found information on inspections for 610 firms from the 2009 data set).

Category	Obs.	Certified	Have knowledge of LEP	Total labor force	College-educated	Inspections	Bribed	Exported
Have knowledge of LEP Paid limited attention to LEP Certified Uncertified	907 1309 369 1847	0.281 0.087	0.647 0.309	14.840 4.688 26.041 5.407	0.314 0.096 0.460 0.130	0.190 0.092 0.251 0.108	0.447 0.207 0.509 0.264	0.066 0.014 0.149 0.012

Table 2: Summary statistics of important explanatory variable across groups

Notes: Summary statistics are calculated for the regression sample of columns (2) and (3) of Table 3, using 2216 observations. This regression sample includes firms in the food and beverage industry that did not change either location or industry of operation over the three years of the sample (2011, 2013 and 2015) and had manufacturing as the main production sector. All variable means are significantly different for firms that had knowledge of the LEP (compared to those that didn't pay attention to it) at the 1% level. Similarly, all variable means are significant different for firms that were certified, versus those that were not certified, at the 1% level.

rate was about 8.7% for firms that paid limited attention to the environmental law, compared to 28% for firms that were aware of it to some extent. On average, the firms that had some awareness were slightly larger (with an average of 15 employees), while firms with limited attention had about 5 employees. Inattentive respondents were less likely to be college-educated (only about 9.6% of them are college-educated), compared to 31.4% of respondents with some awareness of the LEP who were college-educated. Firms where respondents were aware were also located in districts where more inspections had taken place (even though the absolute number is rather low), suggesting that inspections may be a means to bring the law and its requirements to the attention of respondents. Interestingly, the share of respondents having paid bribes was more than double for firms with some knowledge of the LEP (45%), compared to those that paid limited attention to it (21%). Firms where respondents were aware of the LEP were also more likely to export (6.6%, compared to 1.4% for those firms where respondents had paid limited attention to the LEP). Note that these variable means are all significantly different between respondents who had paid limited attention, and those that had some knowledge of the LEP, at the 1% level.

Firms that received the environmental standards certificate were less likely to have owners/managers paying limited attention to the LEP (31%) compared to firms that did not have the certification (65%). Likewise, firms that received certification were likely to be larger (an average of 26 employees, compared to 5 employees at firms that have not received certification). Certified firms were also more likely to have college-educated respondents (46% versus 13%), and more likely to have been inspected in the past as well compared to non-certified firms. They were more likely to have paid bribes (about 51% versus 26% of non-certified firms). Lastly, 15% of firms that received certification exported, whereas only 1.2% of those without certification end up exporting. Again, the variable means are all significantly different between respondents who worked at firms that had been certified, and those that were not, at the 1% level.

The close association between awareness and certification can also be seen in Figure 1, which plots the distribution of the levels of attention paid by respondents to the LEP in our sample over the two categories of certification status, namely firms that have not received a certificate, versus those that have. There appears to be a very clear association between limited attention, and not having received certification (the distribution of non-certified firms is heavily concentrated in the category of those that "have no knowledge of or interest in the LEP"), whereas the difference is more subtle for those firms that have received certification.

In Figure 2, we plot the proportion of respondents that have paid limited attention to the



Figure 1: Association between awareness and certification at the overall level

Notes: Source: UNU-WIDER Vietnam Database. The plot uses data of the regression sample of columns (2) and (3) of Table 3, using 2216 observations. The regression sample includes firms in the food and beverage industry that did not change either location or industry of operation over the three years of the sample (2011, 2013 and 2015) and had manufacturing as the main production sector.

LEP, as well as the environmental certification rates, across provinces in our sample of firms in the food and beverage industry. We see wide heterogeneity in both levels of awareness and certification levels across provinces. For instance, in Phu Tho, we find that both are rather low (6% of the firms have received certification, versus 92.24% were inattentive to the LEP). On the other hand, in Ho Chi Minh City, about 37% of firms had respondents who had paid limited attention to the LEP, and 29% of firms were certified. In general, it is clear that provinces that have higher levels of attentivity/knowledge of the LEP also have higher levels of certification. While these are just correlations, they still provide descriptive evidence on our main hypothesis.

Table 6 in the appendix presents summary statistics on our instrumental variable, namely the index that we created to denote the skill level of the respondent's parents' occupations. This index is generated as the sum of the skill level of the respondent's mothers' and fathers' occupations. Panel A presents summary statistics on these individual indices for the mothers and fathers respectively. The left-hand column presents different categories of occupations, along with the value taken by the index for these occupations in brackets. More skilled jobs take lower values (e.g. the respective indices take the value of "1" if the parent works in a management role), whereas lower skilled jobs take higher values (e.g. the respective indices take the value of "5" if the parent works as an unskilled worker).

We find that a larger share of the respondent's mothers are in higher skill occupations than fathers. About 77% of respondent's mothers are unskilled workers, compared to 84% with fathers who are unskilled workers. Panel B presents the distribution of the sum of the two



Figure 2: Association between awareness and certification at the province-level

Notes: Source: UNU-WIDER Vietnam Database. The plot uses data of the regression sample of columns (2) and (3) of Table 3, using 2216 observations. The regression sample includes firms in the food and beverage industry that did not change either location or industry of operation over the three years of the sample (2011, 2013 and 2015) and had manufacturing as the main production sector.

indices denoting mother's and father's occupations (which we call the index of skill of parents' occupations). Thus, higher levels of this variable denote lower aggregate skill levels of the parents' occupations.

5 Results

5.1 Main Results

Table 3 below presents the main results of this paper. In column (1), we present the results of the estimation using the probit methodology, column (2) includes the results of the recursive bivariate model, whereas column (3) includes the results of the recursive bivariate model using the Mundlak/Chamberlain adjustment to address unobserved heterogeneity. Table 3 presents the coefficients of these estimations. All estimations include province and year fixed effects, a dummy to denote the firms that we identified as needing to obtain the ESC on a mandatory basis, as well as dummies for sub-industries where large firms are required to obtain the ESC, and report robust standard errors. Note that the null hypothesis for zero correlation between the error terms in the first and second stages of the estimations in columns (2) and (3) of Table 3 can be rejected at the 1% level for the model in column (2), and at the 5% level in column (3), using a Wald test (which provides support in favor of the recursive bivariate methodology).

The probit results of column (1) reveal that paying limited attention to the LEP (captured by having no knowledge about it, or no interest in it) is negatively associated with the likelihood of receiving certification for meeting environmental standards. The coefficient is significant at the 1% level. This finding is in line with our intuition. Receiving certification is also positively associated with firm size (firms having a larger workforce are more likely to be certified), and with exporting behavior, which is also expected. Access to information technology is another positive determinant of the likelihood of getting certification, with firms that have internet access and more computers being more likely to be certified, than those that do not have internet access, or less computers. The results of column (1) suggest that college-educated respondents and those for whom profits (or salaries) from the firm are the main source of income are more likely to be working in firms that have received the certification.

Lastly, firms that are household-run enterprises (as opposed to proprietorships, partnerships, limited liability companies, or joint stock companies without state capital) are less likely to receive certification, as are firms that are diversified in their production (i.e. produce more than one good). Household enterprises may face higher costs of obtaining voluntary certification, given that they are often smaller, and pollution control may be expensive for them even if they were aware of the regulation. Secondly, household firms are also less likely to be monitored for meeting mandatory requirements, i.e. the enforcement efforts are likely to be weaker for them. This implies they may not have information on the law, compared to other firms that have been monitored, and thus they may be more inclined towards not obtaining environmental certification.

In columns (2) and (3), we present the results of the second-stage models of the recursive bivariate estimations. We find that paying limited attention to the LEP still has a negative effect on the likelihood of receiving certification in the results of column (2) where we attempt to address endogeneity, and that this variable is again significant at the 1% level. We also confirm some of previous findings about the control variables. For instance, in the results of column (2), we still find that firms that export are more likely to receive certification, as are firms that have more computers. Household firms are less likely to receive certification, as in the results of column (1). Also in line with the results of column (1), respondents who answer that firm profit or income is the main source of income for their household are more likely to be working in firms that have received certification. This is intuitive, and implies that both owners and managers who work primarily at the firm have stronger incentives to ensure that the firm receives certification.

The results of column (3) support our most important finding about the role of being attentive to the LEP on the likelihood of receiving certification, with the variable capturing limited attention of owners/managers having a coefficient of slightly smaller magnitude than in column (2), and retaining significance at the 1% level. Thus, we find that the negative effect of limited attention on the certification likelihood is persistent, even when we attempt to address concerns of unobserved heterogeneity. With the Mundlak/Chamberlain adjustment and the inclusion of averages of time-variant variables, we find that some controls that were significant in previous models are no longer significant in the results of column (3). However, we find that firms that have paid a bribe in the past for whatever reason are less likely to have been certified. One explanation may be that a bribe in this context can be seen as a means to avoid receiving certification. On the other hand, firms may also be paying bribes to expedite receiving certification, with the implication that this result may represent some firms having paid bribes,

Dependent Variable: Whether certified	Probit	Recursive bivariate	Recursive bivariate with Mundlak correction
Column	(1)	(2)	(3)
Paid limited attention to LEP	-0.272	-1.320	-1.235
	(0.083)	(0.355)	(0.411)
Median number of inspections at the district level	-0.108	-0.161	-0.210
	(0.108)	(0.102)	(0.126)
Whether respondent is owner	-0.102	-0.082	0.047
	(0.097)	(0.090)	(0.158)
Whether respondent is male	0.086	0.025	-0.097
Whather firm has cortificate of land use right	(0.060)	(0.077)	(0.152)
Whether firm has certificate of land use fight	(0.032	(0.028	(0.204)
Whether respondent is college educated	0.318	0.156	0.244
	(0.109)	(0.116)	(0.212)
Whether firm has ever paid a bribe	0.028	-0.046	-0.196
	(0.088)	(0.087)	(0.118)
Whether firm exports	0.550	0.461	0.616
	(0.209)	(0.206)	(0.428)
Whether firm produces more than one good	-0.295	-0.216	-0.300
Comment (ouithin district) as alors above	(0.166)	(0.153)	(0.243)
Current (within district) market share	0.0002	0.0002	0.0009
Number of personal computers owned	0.018	0.019	0.077
Number of personal compaters owned	(0.011)	(0.012)	(0.035)
Whether firm faces competition in its field of activity	0.198	0.090	-0.027
	(0.124)	(0.122)	(0.170)
Whether firm had labor disputes in the last two years	0.375	0.292	0.062
	(0.423)	(0.453)	(0.692)
Whether profit/income from firm is the main source of household income	0.329	0.282	0.121
W/kathan firms has multiple summer	(0.129)	(0.121)	(0.175)
whether firm has multiple owners	0.195	0.157	-0.345
Whether firm is a household enterprise	-0.491	-0.362	-0.336
	(0.133)	(0.136)	(0.405)
Total labor force of the firm	0.005	0.003	-0.007
	(0.003)	(0.003)	(0.005)
Age of the respondent	0.0004	-0.001	0.001
	(0.004)	(0.004)	(0.009)
Age of the firm	-0.002	-0.002	0.002
	(0.004)	(0.004)	(0.014)
whether firm has internet access	0.239	0.073	-0.321
	(0.120)	(0.130)	(0.210)
Observations	2174	2216	2216
Wald test-statistic of $rho = 0$		5.076	3.619
P-value		0.024	0.057

Table 3: Main results: second-stage estimations

Notes: Dependent variable is a binary variable denoting whether the firm has been certified for meeting environmental standards. Estimation results of columns (2) and (3) are the second-stage results, corresponding first-stage results are provided in Table 4. All specifications include province and year fixed effects, as well as dummies for sub-sectors where certification is mandatory. The specification in column (3) includes time-averages of all exogenous variables having positive within variation. Regression sample comprises food and beverage industry firms with manufacturing as the main production sector that do not change their location or industry of operation over the duration of the sample. Huber-White standard errors are reported in parentheses. The coefficients of the constant are not reported.

but not yet having received certification. It is difficult to pinpoint the exact channel, in the absence of further information on the reason for which the bribe was paid.

Moreover, firms that belong to districts where more technical inspections have previously taken place are less likely to be certified as well. This suggests that there may be clusters of firms in certain districts that have not been adopting environmental certification, despite some firms having been inspected. In consistency with previous results though, having more computers continues to remain a significant positive determinant of the likelihood of receiving certification for meeting standards.

In Table 4, we present the results of the first-stage estimations corresponding to the results of columns (2) and (3) of Table 3, with column (1) of Table 4 corresponding to the recursive bivariate model, and column (2) to the recursive bivariate model with the Mundlak/Chamberlain adjustment. In the results of column (1), we find that lower skill levels of the parents' occupations

(i.e. higher values of this index) are positively associated with the respondents paying limited attention to the LEP, controlling for respondent-level education. This is in line with our expectation. Moreover, male respondents and college-educated respondents are more likely to be attentive to the LEP, as are older respondents. We also find that firms that have paid a bribe are more likely to have been attentive to the LEP (which, combined with the results of Table 3, suggests that firms may be paying bribes as grease money to receive certifications quicker), as have those with internet access. On the other hand, we find that firms belonging to more competitive sectors are more likely to be knowledgeable of the LEP, as are firms having a larger workforce.

The results of column (2) partially support the results of column (1), with the difference being that fewer control variables are significant once we include the time averages. For instance, our instrumental variable (skill levels of the parents) still has a significant effect on the likelihood of the respondent having been inattentive to the LEP, with the expected sign (respondents whose parents worked in lower skill jobs were more likely to lack knowledge of the LEP). The result on bribery still persists, with firms that bribed being more likely to have paid some attention to the LEP, as is the case for firms that operate under competition. We also find that firms that have multiple owners were more likely to be have been aware of the LEP, which is also somewhat intuitive. Lastly, the previous result on the negative effect of internet access on the likelihood of respondents having limited attention is also confirmed in column (2).

Table 5 presents the marginal effects for some important explanatory variables drawn from the three estimation results presented in Table 3. Note that the marginal effects for the explanatory variables in columns (2) and (3) represent the direct effect of these variables on the likelihood of being certified. We find that firms where respondents had paid limited attention to the LEP were 5 percentage points less likely to obtain certification for meeting environmental standards, than firms where respondents had some knowledge or interest in the law, according to the results of the marginal effects for the probit estimation in column (1).

The size of this effect increases once we control for endogeneity in column (2), where we find that limited attention of owners/managers to the LEP resulted in firms being 33 percentage points less likely to obtain certification for meeting environmental standards, than firms where respondents had some knowledge of it. In column (3), we find the magnitude of this marginal effect to be 30 percentage points, once we include the time averages. Thus, we show that controlling for endogeneity may be significant in this context. Moreover, while the use of the recursive bivariate framework can enable us to interpret these effects as causal, given that we were not able to achieve convergence using this methodology in certain estimations involving a specific sub-sector (refer to Footnote 11), we choose to be conservative and interpret them as providing suggestive evidence on the role of limited attention in determining certification.

We also list the marginal effects for some other important controls in Table 5. We find that in line with the results on the coefficients presented in Table 3, firms with college-educated respondents (owners or managers) and firms that export, are 6 percentage points and 10 percentage points more likely, respectively, to be certified for meeting environmental standards (column (1)). Owning one more computer is also positively associated with the firm receiving certification (an additional computer raises the likelihood of receiving certification by 1.3 percentage points, according to column (3)), while household enterprises are less likely to be certified (6.4 percentage points less likely, according to column (2)). Lastly, firms that belong to districts where more technical inspections took place in previous years are 3.5 percentage

Dependent Variable: Whether aware of LEP Column	Recursive bivariate (1)	Recursive bivariate with Mundlak correction (2))
Index of extent of an elilled a compation of a constant	0.040	0.047
index of extent of unskilled occupations of parents	0.049	0.047
Madian number of increations at the district local	(0.025)	0.025
iviedian number of inspections at the district level	-0.143	-0.030
W/hathay yoon and ant is surrow	(0.095)	(0.114)
whether respondent is owner	0.040	-0.026
W/hathay yoon and ant is male	(0.079)	(0.133)
whether respondent is male	-0.208	-0.137
M/Lather from her sout franks of land one winds	(0.003)	(0.128)
whether firm has certificate of land use right	-0.052	-0.039
M/Lathen warman dant is callene advected	(0.082)	(0.103)
whether respondent is college educated	-0.324	-0.118
	(0.094)	(0.180)
Whether firm has ever paid a bribe	-0.214	-0.240
	(0.072)	(0.098)
Whether firm exports	0.153	-0.099
	(0.243)	(0.449)
Whether firm produces more than one good	0.059	0.222
	(0.124)	(0.197)
Current (within district) market share	0.0007	-0.0002
	(0.0008)	(0.001)
Number of personal computers owned	0.003	0.003
	(0.012)	(0.014)
Whether firm faces competition in its field of activity	-0.356	-0.325
	(0.091)	(0.130)
Whether firm had labor disputes in the last two years	-0.229	-1.356
	(0.773)	(0.908)
Whether profit/income from firm is the main source of household income	-0.058	-0.080
	(0.086)	(0.137)
Whether firm has multiple owners	-0.095	-0.531
	(0.158)	(0.284)
Whether firm is a household enterprise	0.086	0.179
	(0.129)	(0.399)
Total labor force of the firm	-0.013	-0.006
	(0.003)	(0.007)
Age of the respondent	-0.007	-0.0008
	(0.003)	(0.008)
Age of the firm	-0.001	0.005
	(0.003)	(0.015)
Whether firm has internet access	-0.352	-0.345
	(0.106)	(0.203)
Observations	2216	2216

Table 4: Main results: first-stage estimations

Notes: Dependent variable is a binary variable denoting whether the respondent paid limited attention to the LEP. Estimation results are the first-stage results, corresponding second-stage results are provided in columns (2) and (3) of Table 3. All specifications include province and year fixed effects, as well as dummies for sub-sectors where certification is mandatory. Specification in column (2) includes time-averages of all exogenous variables having positive within variation. Regression sample comprises food and beverage industry firms with manufacturing as the main production sector that do not change their location or industry of operation over the duration of the sample. Huber-White standard errors, are reported in parentheses. The coefficients of the constant are not reported.

Dependent Variable: Whether certified Column	Probit (1)	Recursive bivariate (2)	Recursive bivariate with Mundlak correction (3)
Paid limited attention to LEP	-0.050	-0.332	-0.297
	(0.015)	(0.135)	(0.145)
Whether respondent is college educated	0.058	0.028	0.041
	(0.020)	(0.019)	(0.035)
Whether firm exports	0.100	0.082	0.103
	(0.038)	(0.037)	(0.071)
Number of personal computers owned	0.003	0.003	0.013
	(0.002)	(0.002)	(0.006)
Whether firm is a household enterprise	-0.089	-0.064	-0.056
	(0.024)	(0.022)	(0.066)
Median number of inspections at the district level	-0.020	-0.029	-0.035
	(0.020)	(0.019)	(0.022)
Observations	2174	2216	2216

Table 5: Marginal effects of important explanatory variables

Notes: Dependent variable is a binary variable denoting whether the firm has been certified for meeting environmental standards. Marginal effects for some important explanatory variables (calculated at the dydx level) are presented, corresponding to the second-stage coefficients of Table 3. Marginal effects are calculated at means. Bootstrapping is used to calculate the direct marginal effect for the "Paid limited attention to LEP" variable in columns (2) and (3), run over 1000 iterations. All specifications include province and year fixed effects, as well as dummies for sub-sectors where certification is mandatory. Specification in column (3) includes time-averages of all exogenous variables having positive within variation. Regression sample comprises food and beverage industry firms with manufacturing as the main production sector that do not change their location or industry of operation over the duration of the sample. Huber-White standard errors are reported in parentheses. *,** and *** respectively denote significance at 10%, 5% and 1% levels. The coefficients of the constant are not reported.

points less likely to be certified themselves (as suggested in column (3)).

These results confirm our hypothesis regarding the importance of paying attention to the laws on whether a firm receives voluntary certification for meeting environmental standards. In the next subsection, we derive marginal effects over important subgroups of the data sample, to examine heterogeneities in the magnitude of the main effect.

5.2 Extensions

In Table 7 in the Appendix, we present some additional results based on our main estimation results of Table 3. We estimate whether the marginal effect of limited attention on certification for meeting environmental standards varies along different dimensions (such as province, time, education, etc.). We use the recursive bivariate probit estimation of column (3) of Table 3 to compute these heterogeneous marginal effects, given that it mitigates the concern of endogeneity, and it also attempts to address the problem of unobserved heterogeneity through the Mundlak/Chamberlain adjustment.

We find that the absolute value of the marginal effect of limited attention on the likelihood of certification is higher for firms with college-educated respondents, and those that export, i.e. firms that exported, or had college-educated owners or managers, were more likely to have obtained the voluntary certification if they had paid attention to the LEP, than firms that didn't have college-educated respondents, or didn't export. This confirms findings from previous studies on the role of education and exporting behavior in determining the adoption of voluntary certificates (Ervin et al. 2012; Tambunlertchai et al. 2013). We also find that negative effect of limited attention on certification is weaker for household enterprises, i.e.

household enterprises with owners or managers who were aware of the LEP were less likely to obtain certification than non-household enterprises. In line with our previous results, this may be due to household enterprises facing larger costs for adopting voluntary standards, or due to weaker enforcement efforts towards them (that may be a potential source of information for firms on the LEP).

Interestingly, we find that firms that have made informal payments, or bribed, are more likely to obtain certification if they are attentive to the LEP, than firms that have not bribed (based on the absolute values of the marginal effects). This provides further support in favour of our finding that paying bribes may be a means to push authorities to issue certificates for meeting environmental standards (rather than avoid getting the standards) to firms, and thus that the cost of getting certified may be higher for firms that have never bribed, i.e. it may be more difficult for them to obtain certification, even if they are aware of the LEP.

Finally, we find that the absolute value of the marginal effect of limited attention on the likelihood of certification is higher for firms that have been inspected at least once, i.e. firms that had been inspected at least once were more likely to obtain certification if they were attentive to the LEP (compared to firms that had never been inspected). This suggests that enforcement may be a suitable policy mechanism to both increase awareness of the LEP, but also increase the certification likelihood.

The value of the marginal effect of limited attention on the likelihood of being certified also varies across provinces, with the highest (negative) marginal effects observed in Hai Phong, Ho Chi Minh City (HCMC) and in Long An, which implies that firms in these provinces with owners or managers who were attentive to the LEP were more likely to have obtained environmental certification, than in other provinces. Moreover, we also find that the marginal effect varies over time, with the strongest effect observed in 2011 (firms with limited attention were 33 percentage points less likely to obtain environmental certification then), which declined to about 10 percentage points by 2015. As intuition suggests, with time, the effect of knowledge of the LEP on obtaining voluntary certification should weaken, as this finding suggests.

6 Conclusion and Policy Implications

Our study attempts to shed light on the role of limited attention paid by owners and managers of firms to the Law on Environmental Protection, on the likelihood of voluntarily obtaining certification for meeting environmental standards in the food and beverage industry in Vietnam. We find that firms where owners and managers paid limited attention to the law were 30 percentage points less likely to be certified for meeting environmental standards, while controlling for measures of enforcement which have traditionally been found to be important in the economic literature, and accounting for firms that needed to obtain certification (on a mandatory basis).

Moreover, firms where the owners or managers are college-educated, that export goods, and that have previously been inspected for technical reasons such as environmental violations, are more likely to be certified if they have been attentive to the LEP, suggesting that education, international linkages, and traditional enforcement mechanisms like monitoring may be playing an important role in incentivizing firms to get certified in this context (in line with the previous literature). This effect, however, is weaker for firms that are household enterprises, because the

cost of compliance may be higher for them (and levels of enforcement are also likely be lower). Interestingly, we also find that firms that have bribed are more likely to obtain certification if their owners/managers had been attentive to the LEP (as opposed to firms that had never bribed). Thus, bribing may be seen as a form of grease money payment in this particular context, in line with other studies (Bai et al., 2019).

The results also highlight which segment of the population is more likely to be have paid attention to the regulations. For instance, given that firms with internet access are more likely to be aware of regulations, policy-makers may benefit from targeting information and education campaigns to those firms that lack internet access, to owners of businesses who are not very educated, or to household enterprises. Our results also suggest that firms with multiple owners, or those that faced competition from other firms in their field of activity, were also more likely to be have knowledge of the LEP, as were larger firms. Thus, smaller firms may need further incentives to adopt voluntary standards in the absence of access to information on them.

The key takeaway from our study is that addressing the behavioral anomaly of limited attention, and thereby ensuring that firm owners and managers have knowledge of the laws and understand them, may be a critical determinant of ensuring that firms participate in voluntary programs, especially in contexts where institutions are weak and factors such as corruption may incentivize violations on part of firms.

Our results have important policy implications in the context of Vietnam. Given that the food and beverage industry has the highest pollution load in Vietnam (Dore et al., 2008), the findings of this study are very relevant for policy-makers looking for low-cost solutions to ensure greater environmental compliance. While traditional enforcement and implementation tools remain important, bringing to salience information on laws and regulations should be a complementary policy measure to ensure that firms have no information constraints to abide with laws, even for inducing the voluntary adoption of policies such as certificates or environmental standards. Of course, it remains to be seen whether these results are applicable to firms belonging to other industries.

Somanathan (2010) highlights the importance of using information policies along with regulations to improve environmental quality in developing countries. While firms may continue evading adoption of policy instruments if the benefits of doing so exceed the costs, bringing laws and their requirements to salience has the potential, as our study shows, to make it more economically viable for firms to obtain voluntary certificates, by relaxing their information constraints.

Acknowledgements

We are grateful to Ho Hoang Anh, Do Huu Luat and the Vietnam Law and Legal Forum for help in understanding environmental law in Vietnam. We are also grateful to Armin Burgmeier and Neda Trifkovic for help with the data. We are responsible for all omissions and errors.

References

- Arimura, T. H., Hibiki, A., and Katayama, H. (2008). Is a Voluntary Approach an Effective Environmental Policy Instrument?. A Case for Environmental Management Systems. *Journal* of Environmental Economics and Management, 55(3):281–295.
- Arora, S. and Gangopadhyay, S. (1995). Toward a Theoretical Model of Voluntary Overcompliance. *Journal of Economic Behavior and Organization*, 28(3):289–309.
- Bai, J., Jayachandran, S., Malesky, E. J., and Olken, B. A. (2019). Firm growth and corruption: Empirical evidence from Vietnam. *Economic Journal*, 129(618):651–677.
- Banerjee, A., Banerji, R., Duflo, E., Glennerster, R., and Khemani, S. (2008). Pitfalls of participatory programs: evidence from a randomized evaluation in education in India. *American Economic Journal: Economic Policy*, 2(1):1–30.
- Banerjee, A., Hanna, R., and Mullainathan, S. (2013). Corruption. In Gibbons, R. and Roberts, J., editors, *Handbook of Organizational Economics*, pages 1109–47. Princeton: Princeton University Press.
- Beaman, L., Magruder, J., and Robinson, J. (2014). Minding Small Change Among Small Firms in Kenya. *Journal of Development Economics*, 108:69–86.
- Blackman, A. (2010). Alternative pollution control policies in developing countries. *Review of Environmental Economics and Policy*, 4(2):234–253.
- Blackman, A. and Guerrero, S. (2012). What Drives Voluntary Eco-certification in Mexico? *Journal of Comparative Economics*, 40(12):256–268.
- Blackman, A., Lahiri, B., Pizer, W., Rivera Planter, M., and Muñoz Piña, C. (2010). Voluntary Environmental Regulation in Developing Countries: Mexico's Clean Industry Program. *Journal* of Environmental Economics and Management, 60(3):182–192.
- Bordalo, P., Gennaioli, N., and Shleifer, A. (2013). Salience and Consumer Choice. *Journal of Political Economy*, 121(5):803–843.
- Cerruti, D., Daminato, C., and Filippini, M. (2019). The impact of policy awareness: Evidence from vehicle choices response to fiscal incentives. *Economics Working Paper Series*, 19/316.
- Chamberlain, G. (1982). Multivariate regression models for panel data. *Journal of Econometrics*, 18(1):5–46.
- Chetty, R., Looney, A., and Kroft, K. (2009). Salience and Taxation: Theory and Evidence. *American Economic Review*, 99(4):1145–1177.
- Christmann, P. and Taylor, G. (2001). Globalization and the Environment: Determinants of Firm Self-Regulation in China. *Journal of International Business Studies*, 32(3):439–458.
- Decker, C. S. (2003). Corporate Environmentalism and Environmental Statutory Permitting. *Journal of Law and Economics*, 46(1):103–129.
- Dellavigna, S. (2009). Psychology and Economics: Evidence from the Field. *Journal of Economic Literature*, 47(2):315–372.

- Dore, G., Brylski, P., Nygard, J., Thi, P., and Tran, T. (2008). Review and analysis of the pollution impacts from the Vietnamese manufacturing sectors. *EASRE, The World Bank*.
- Duflo, E., Hanna, R., and Ryan, S. P. (2012). Incentives work: getting teachers to come to school. American Economic Review, 102(4):1241–1278.
- Ervin, D., Wu, J., Khanna, M., Jones, C., and Wirkkala, T. (2012). Motivations and Barriers to Corporate Environmental Management. *Business Strategy and the Environment*, 22(6):390– 409.
- Eskeland, G. S. and Jimenez, E. (1992). Policy instruments for pollution control in developing countries. *World Bank Research Observer*, 7(2):145–169.
- Gabaix, X. and Laibson, D. (2006). Shrouded Attributes, Consumer Myopia, and Information Suppression in Competitive Markets. *Quarterly Journal of Economics*, 121(2):505–540.
- General Statistics Office (2015). Statistical Yearbook of Vietnam 2015. http://www.gso.gov.vn/default_en.aspx?tabid=515&idmid=&ItemID=16052.
- Greene, W. (2008). Discrete Choice Modeling. In Mills, T. and Patterson, K., editors, *The Handbook of Econometrics: Vol. 2, Applied Econometrics*. Palgrave, London.
- Greene, W. (2018). Econometric Analysis. Pearson Education, 8th edition.
- Greene, W. and Zhang, Q. (2019). Nonlinear and Related Panel Data Models. In Tsionas, M., editor, *Panel Data Econometrics*. Elsevier.
- Han, S. and Vytlacil, E. J. J. (2017). Identification in a Generalization of Bivariate Probit Models with Dummy Endogenous Regressors. *Journal of Econometrics*, 199(1):63–73.
- Hanna, R., Mullainathan, S., and Schwartzstein, J. (2014). Learning Through Noticing: Theory and Evidence from a Field Experiment. *Quarterly Journal of Economics*, 129(3):1311–1353.
- Hirshleifer, D. and Teoh, S. H. (2003). Limited Attention, Information Disclosure, and Financial Reporting. *Journal of Accounting and Economics*, 36(1-3):337–386.
- Hossain, T. and Morgan, J. (2006). ...Plus Shipping and Handling: Revenue (non) Equivalence in Field Experiments on eBay. *Advances in Economic Analysis and Policy*, 6(2):63–91.
- Innes, R. and Sam, A. G. (2008). Voluntary Pollution Reductions and the Enforcement of Environmental Law: An Empirical Study of the 33/50 Program. *Journal of Law and Economics*, 51(2):271–296.
- International Trade Administration (2019). Environmental and Pollution Control Equipment and Services. *Vietnam Commercial Guide*.
- IPCC (2014). Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.
- IQAir (2019). World's Most Polluted Countries in 2019 PM2.5 Ranking. https://www.iqair.com/world-most-polluted-countries.

- Karlan, D., McConnell, M., Mullainathan, S., and Zinman, J. (2016). Getting to the Top of Mind: How Reminders Increase Saving. *Management Science*, 62(12):3393–3411.
- Khanna, M. and Liao, Y. (2014). Globalization and Voluntary Environmental Management in Developing Countries. *Frontiers of Economics in China*, 9(1):138–163.
- Khanna, M. and Speir, C. (2013). Motivations for Proactive Environmental Management. *Sustainability (Switzerland)*, 5(6):2664–2692.
- King, A. A., Lenox, M. J., and Terlaak, A. (2005). The Strategic Use of Decentralized Institutions: Exploring Certification with the ISO 14001 Management Standard. Academy of Management Journal, 48(6):1091–1106.
- Koszegi, B. and Szeidl, A. (2013). A Model of Focusing in Economic Choice. *Quarterly Journal* of *Economics*, 128(1):53–104.
- Maddala, G. S. (1986). Limited-dependent and Qualitative Variables in Econometrics. Cambridge University Press.
- Montiel, I. and Husted, B. W. (2009). The Adoption of Voluntary Environmental Management Programs in Mexico: First Movers as Institutional Entrepreneurs. *Journal of Business Ethics*, 88(SUPPL. 2):349–363.
- Mourifié, I. and Méango, R. (2014). A Note on the Identification in Two Equations Probit Model with Dummy Endogenous Regressor. *Economics Letters*, 125(3):360–363.
- Mundlak, Y. (1978). On the pooling of time series and cross section data. *Econometrica*, 46(1):69–85.
- Nielsen (2015). Sustainability Influences Purchase Intent of Vietnamese Conusmers. *Corporate Sustainability Report*.
- Nkonya, E., Pender, J., and Kato, E. (2008). Who knows, who cares? The determinants of enactment, awareness, and compliance with community natural resource management regulations in Uganda. *Environment and Development Economics*, 13(1):79–101.
- Potoski, M. and Prakash, A. (2005). Green clubs and voluntary governance: ISO 14001 and firms' regulatory compliance. *American Journal of Political Science*, 49(2):235–248.
- Prime Minister's Office (2018). Circular No. 27/2018/QD-TTg: Promulgating Vietnam Standard Industrial Classification.
- Segerson, K. and Miceli, T. J. (1998). Voluntary environmental agreements: Good or bad news for environmental protection? *Journal of Environmental Economics and Management*, 36(2):109–130.
- Sharma, S. and Tarp, F. (2018). Does Managerial Personality Matter? Evidence from Firms in Vietnam. *Journal of Economic Behavior and Organization*, 150:432–445.
- Somanathan, E. (2010). Effects of information on environmental quality in developing countries. *Review of Environmental Economics and Policy*, 4(2):275–292.

- Tambunlertchai, K., Kontoleon, A., and Khanna, M. (2013). Assessing participation in voluntary environmental programmes in the developing world: The role of FDI and export orientation on ISO14001 adoption in Thailand. *Applied Economics*, 45(15):2039–2048.
- The Government of Vietnam (2015). Circular No. 18/2015/ND-CP: Prescribing Environmental Protection Master Plan, Strategic Environmental Assessment, Environmental Impact Assessment and Environmental Protection Plan.
- United Nations University UNU-WIDER (2011). Viet Nam SME database.
- Videras, J. and Alberini, A. (2000). The Appeal of Voluntary Environmental Programs: Which Firms Participate and Why? *Contemporary Economic Policy*, 18(4):449–460.
- Vidovic, M. and Khanna, N. (2007). Can Voluntary Pollution Prevention Programs Fulfill Their Promises? Further Evidence from the EPA's 33/50 Program. *Journal of Environmental Economics and Management*, 53(2):180–195.
- Vietnam Law and Legal Forum (2008). Circular No. 05/2008/TT-BTNMT of December 8, 2008, Guiding Strategic Environmental Assessment, Environmental Impact Assessment and Environmental Protection Commitment.
- Vincent, J. R. (2010). Microeconomic analysis of innovative environmental programs in developing countries. *Review of Environmental Economics and Policy*, 4(2):221–233.
- Wendling, Z. A., Emerson, J. W., Esty, D. C., Levy, M. A., and et al. de Sherbinin, A. (2018). 2018 Environmental Performance Index. New Haven, CT: Yale Center for Environmental Law & Policy.
- Wilde, J. (2000). Identification of Multiple Equation Probit Models with Endogenous Dummy Regressors. *Economics Letters*, 69(3):309–312.
- Wooldridge, J. M. (2005). Unobserved Heterogeneity and Estimation of Average Partial E. In *Identification and Inference for Econometric Models: Essays in Honor of Thomas Rothenberg*, pages 27–55. Cambridge University Press.
- Wooldridge, J. M. (2010). Econometric Analysis of Cross Section and Panel Data. MIT Press.
- World Bank (2010). World Development Report 2010 : Development and Climate Change.

World Bank (2019). Vietnam Overview. https://www.worldbank.org/en/country/vietnam/overview.

Appendix

Panel A: Index of skills of mother and father's occupations						
No of obs. by category	Mother	Father				
Management (1)	46	16				
Higher educated professional worker (2)	26	14				
Medium educated professional worker (3)	46	23				
Semi-skilled workers (4)	400	303				
Unskilled workers (5)	1694	1854				
Total observations	2212	2210				
Panel B: Sum of indices of mother's and father's occupations						
Sum of indices	Observations					
2	ç)				
3	4					
4	13					
5	25					
6	43					
7	29					
8	277					
9	143					
10	1673					
Total observations	22	07				

Table 6: Distribution of the index of skills of parent's occupations

Notes: Semi-skilled workers include office and customer service workers, sales and security workers, agricultural workers, production workers, operator and installation workers, and members of the army. In case the value for one of the two individual indices in Panel A is missing, the sum of the the indices of parents' occupations takes the value of the non-missing index. Distributions are derived for the regression sample of columns (2) and (3) of Table 3. This regression sample includes firms in the food and beverage industry that did not change either location or industry of operation over the three years of the sample (2011, 2013 and 2015) and had manufacturing as the main production sector.

Binary variables	Provin	ice Year		lear		
Whether firm exports	No	-0.187 (0.093)	Ha Noi	-0.298 (0.132)	2011	-0.325 (0.133)
	Yes	-0.388	Phu Tho	-0.018	2013	-0.241
Whether respondent is college-educated	No	-0.151	Ha Tay	-0.170	2015	-0.099
	Yes	(0.080) -0.462	Hai Phong	(0.093) -0.414		(0.061)
Whether firm is a household enterprise	No	(0.161) -0.492	Nghe An	(0.155) -0.122		
	Yes	(0.164) -0.136	Quang Nam	(0.072) -0.186		
Whether firm paid a bribe	No	(0.075)	Khanh Hoa	(0.095)		
Whether fifth paid a bride	NO	-0.145 (0.078)		(0.130)		
	Yes	-0.370 (0.144)	Lam Dong	-0.347 (0.147)		
Whether firm was inspected	No	-0.136	НСМС	-0.400		
	Yes	-0.286 (0.123)	Long An	(0.101) -0.384 (0.148)		

Table 7: Heterogeneous marginal effects

Notes: Dependent variable is a binary variable denoting whether the firm has been certified for meeting environmental standards. Marginal effect of awareness on being certified is calculated over some binary variables, province, and year (at the dydx level), corresponding to the second-stage coefficient of column (3) of Table 3, using 2216 observations. Marginal effects are calculated at means. Specification of column (3) of Table 3 includes province and year fixed effects, dummies for sub-sectors where certification is mandatory, and time-averages of all exogenous variables having positive within variation. Regression sample comprises food and beverage industry firms with manufacturing as the main production sector that do not change their location or industry of operation over the duration of the sample. Huber-White standard errors are reported in parentheses. The coefficients of the constant are not reported.

Working Papers of the Center of Economic Research at ETH Zurich

- (PDF-files of the Working Papers can be downloaded at www.cer.ethz.ch/research/working-papers.html).
- 20/338 M. Filippini, S. Srinivasan Voluntary adoption of environmental standards and limited attention: Evidence from the food and beverage industry in Vietnam
- 20/337 F. Böser, C. Colesanti Senni Emission-based Interest Rates and the Transition to a Low-carbon Economy
- 20/336 L. Bretschger, E. Grieg, P. J.J. Welfens, T. Xiong Corona Fatality Development and the Environment: Empirical Evidence for OECD Countries
- 20/335 M. Arvaniti, W. Habla The Political Economy of Negotiating International Carbon Markets
- 20/334 N. Boogen, C. Daminato, M. Filippini, A. Obrist Can Information about Energy Costs Affect Consumers Choices? Evidence from a Field Experiment
- 20/333 M. Filippini, N. Kumar, S. Srinivasan Nudging the Adoption of Fuel-Efficient Vehicles: Evidence from a Stated Choice Experiment in Nepal
- 20/332 L. Bretschger, E. Grieg Exiting the fossil world: The effects of fuel taxation in the UK
- 20/331 H. Gersbach, E. Komarov Research Bubbles
- 20/330 E. V. Dioikitopoulos, C. Karydas Sustainability traps: patience and innovation
- 19/329 M. Arvaniti, C. K. Krishnamurthy, A. Crepin Time-consistent resource management with regime shifts
- 19/328 L. Bretschger, K. Pittel Twenty Key Questions in Environmental and Resource Economics
- 19/327 C. Karydas, A. Xepapadeas Climate change financial risks: pricing and portfolio allocation
- 19/326 M. Filippini, S. Srinivasan Investments in Worker Health and Labor Productivity: Evidence from Vietnam

- 19/325 H. Gersbach Democratizing Tech Giants! A Roadmap
- 19/324 A. Brausmann, M. Flubacher and F. Lechthaler Valuing meteorological services in resource-constrained settings: Application to smallholder farmers in the Peruvian Altiplano
- 19/323 C. Devaux and J. Nicolai Designing an EU Ship Recycling Licence: A Roadmap
- 19/322 H. Gersbach Flexible Majority Rules for Cryptocurrency Issuance
- 19/321 K. Gillingham, S. Houde and A. van Benthem Consumer Myopia in Vehicle Purchases: Evidence from a Natural Experiment
- 19/320 L. Bretschger Malthus in the Light of Climate Change
- 19/319 J. Ing and J. Nicolai Dirty versus Clean Firms' Relocation under International Trade and Imperfect Competition
- 19/318 J. Ing and J. Nicolai North-South diffusion of climate-mitigation technologies: The crowding-out effect on relocation
- 19/317 J. Abrell, M. Kosch and S. Rausch How Effective Was the UK Carbon Tax? - A Machine Learning Approach to Policy Evaluation
- 19/316 D. Cerruti, C. Daminato and M. Filippini The Impact of Policy Awareness: Evidence from Vehicle Choices Response to Fiscal Incentives
- 19/315 M. Filippini, N. Kumar and S. Srinivasan Energy-related financial literacy and bounded rationality in appliance replacement attitudes: Evidence from Nepal
- 19/314 S. Houde and E. Myers Heterogeneous (Mis-) Perceptions of Energy Costs: Implications for Measurement and Policy Design
- 19/313 S. Houde and E. Myers Are Consumers Attentive to Local Energy Costs? Evidence from the Appliance Market

19/312 N. Kumar

A model-based clustering approach for analyzing energy-related financial literacy and its determinants

- 19/311 C. Karydas and A. Xepapadeas Pricing climate change risks: CAPM with rare disasters and stochastic probabilities
- 19/310 J.Abrell,S. Rausch and C. Streitberger Buffering Volatility: Storage Investments and Technology-Specific Renewable Energy Support
- 19/309 V. Britz Negotiating with frictions
- 19/308 H. Gersbach and S. Papageorgiou On Banking Regulation and Lobbying
- 18/307 V. Britz, A. Ebrahimi and H. Gersbach Incentive Pay for Policy-makers?
- 18/306 C. Colesanti Senni and N. Reidt Transport policies in a two-sided market
- 18/305 A. Schäfer and A. Stünzi The impact of green preferences on the relevance of history versus expectations
- 18/304 M. Filippini and S. Srinivasan Impact of religious participation, social interactions and globalisation on meat consumption: evidence from India
- 18/303 H. Gersbach and M.-C. Riekhof Permit Markets, Carbon Prices and the Creation of Innovation Clusters
- 18/302 M. Hersche and E. Moor Identification of Causal Intensive Margin Effects by Difference-in-Difference Methods
- 18/301 L. Kleemann and M.-C. Riekhof Changing background risk and risk-taking - Evidence from the field
- 18/300 J. Blasch and C. Daminato Behavioral anomalies and energy-related individual choices: the role of status-quo bias
- 18/299 S. Rausch and H. Schwerin Does Higher Energy Efficiency Lower Economy-Wide Energy Use?
- 18/298 H. Gersbach, U. Schetter and M. Schneider Economic Rationales for Investments in Science